Author’s response to reviews

Title: Umbilical cord bilirubin as a predictor of neonatal jaundice: a retrospective cohort study

Authors:
Kelsey Jones (kelseyjones@gmail.com)
Sophie Grossman (sophie.grossman8@gmail.com)
Dharshini Kumaranayakam (dharshinikumar@hotmail.com)
Arati Rao (aratirao@nhs.net)
Greg Fegan (g.w.fegan@swansea.ac.uk)
Narendra Aladangady (narendra.aladangady@homerton.nhs.uk)

Version: 2 Date: 01 Aug 2017

Author’s response to reviews:

Changes to the ethical approval declaration section have been made as requested.